

Received April 4, 2022, accepted April 24, 2022, date of publication May 3, 2022, date of current version May 12, 2022. *Digital Object Identifier* 10.1109/ACCESS.2022.3172285

Large Event Halls Evacuation Using an Agent-Based Modeling Approach

LIVIU-ADRIAN COTFAS^{®1}, CAMELIA DELCEA^{®1}, LIVIA-DIANA IANCU¹, CORINA IOANĂȘ², AND CRISTINA PONSIGLIONE^{®3}

¹Department of Economic Informatics and Cybernetics, Bucharest University of Economic Studies, 010552 Bucharest, Romania
 ²Department of Accounting and Audit, Bucharest University of Economic Studies, 010552 Bucharest, Romania
 ³Department of Industrial Engineering, University of Naples Federico II, 80125 Naples, Italy

Corresponding author: Camelia Delcea (camelia.delcea@softscape.ro)

ABSTRACT The paper explores the usage of agent-based modeling in the context of large event halls evacuation during music festivals and cultural events. An agent-based model is created in NetLogo 6.2.2 for better representing the human behavior when involved in such situations. A series of characteristics have been set for the agents in order to preserve their heterogeneity in terms of speed, age, locomotion impairment, familiarity with the environment, evacuating with another person, choosing the closest exit or not, selecting the closest path to the exits. An "adapted cone exit" approach has been proposed in the paper in order to facilitate the guidance of the agents in the agent-based model to the closest exit and its advantages have been proved in comparison with the classical "cone exit" approach. Different evacuation scenarios have been simulated and analyzed for better observing the capabilities of evacuation modeling in the case of evacuation emergencies. Besides the overall evacuation time, an average evacuation time has been determined for the agents based on the individual evacuation time, which can be easily connected with a risk indicator associated to each situation. Due to the visual interface offered by the agent-based model, coupled with the evacuation indicators, the proposed model can allow the identification of the main factors that may contribute to a prolonged evacuation process (e.g. overcrowding at one of the exits, not choosing the appropriate door, evacuating with a friend/parent) and the potential measures to be considered for insuring a safe evacuation process.

INDEX TERMS Agent-based modeling, large event halls, evacuation process, simulation.

I. INTRODUCTION

Evacuation in a timely manner and in compliance with safety regulations is of great importance when it comes to an unforeseen situation [1]. In general, emergency evacuation can be considered as a traditional problem of identifying the optimal route. Numerous classical algorithms have been proposed to solve this problem, for example the Depth-First-Search algorithm [2], dynamic programming [3], Dijkstra's algorithm [4] and ant colony optimization algorithms [5]. However, evacuating a mass of people is a complicated process. The algorithms mentioned above do not take into account psychological factors, particularities of the population (gender, age, locomotion impairments) and interpersonal

The associate editor coordinating the review of this manuscript and approving it for publication was Yichuan Jiang¹⁰.

relationships. All these factors have a decisive impact on evacuation [6], [7].

Currently, simulations are an effective way to estimate evacuation times under the influence of variable and invariable factors. In the field of emergency evacuations, researchers most often use virtual models to simulate such evacuations. Isobe *et al.* [8] tested a walking situation and determined that human behavior differed depending on the environment. Yang *et al.* [9] through computer simulations found out that a point of interest consists in the stairs of buildings, because in case of emergency evacuation, that is the place where the traffic of people is congested. Helbing and Molnar's social model [10] and Blue and Adler's automatic model [11] have often been applied in evacuation simulations. The real-world simulation of an emergency evacuation allows the recording of the entire evacuation procedure. However, the current studies are limited by the environment characteristics and factors that are difficult to implement, such as the multitude of behaviors that people could manifest in such a situation, unforeseen events that obstruct the evacuation process and the diversity of the evacuees. In most of the cases the simulations are limited to the evacuation of some rooms/interior buildings.

Emergency evacuation of a location during an event is of major importance due to the large number of persons participating, in general, to the event. The simulation of the large spaces evacuation in dangerous conditions is a necessary measure to prevent or to reduce the number of victims [3]. If, when exposed to a source of danger, the participants at an event are not evacuated effectively, serious consequences may arise as a result. For example, in the tragedy of October 30, 2015 in Club Colectiv,¹ located in Bucharest, Romania, a place where after a fire broke out, 64 people died and another 186 people have been injured. This incident represents the worst fire in Romania in a nightclub and the worst accident in the country in recent decades. In order to prevent such accidents, emergency evacuation planning is essential in carrying out such an activity.

Considering the other approaches from the public spaces evacuation scientific literature, it can be observed that in some of these approaches the evacuation population is regarded as a single homogenous population.

As a result, the emergency evacuation of a large number of people from a location, in the shortest possible time and at the highest possible level of safety, is extremely important. In this context, the present study uses an agent-based model created in NetLogo 6.2.2 for simulating the evacuation process from a large event tent in order to highlight the evacuation times and to find the potential issues that might appear during the evacuation. The use of an agent-based approach for modeling the evacuation population from large event halls is intended to overcome the gaps in the literature which derive from the assumptions made in other approaches. For example, one of the assumptions in the social force models is that the population is homogenous, not accounting for the individual characteristics or behavior of the people involved in such an event. On the other hand, the use of cellular automata models, which are able to account for these characteristics, conducts to complex models, hard to implement and run due to the existence of only one type of agent that should possess at the same time the characteristics of the evacuation population and of the environment. Even more, in the case of cellular automata models, as the evacuation agents are represented by fixed pieces of ground, the movement can be made only from one piece of ground to the other - in this manner, an agent is not able to initiate a movement of a lengths smaller than the size of a patch. This limitation of the cellular automata models makes the models using this approach to move away from the real-life evacuation scenarios. All these limitations can be overpassed in an agent-based model. As a result, the use of agent-based modeling for such evacuation situations comes easier as the population can be constructed as heterogenous and, by providing a series of agents, in the agent-based models, the design of the environment, the interactions among evacuation population and the individual rules of movement and characteristics of the evacuees can easily be modeled. In terms of movement, as the evacuation agents are not modeled as part of the ground, but as moving agents on top of the ground, the evacuation agents can stop at any position, not being forced to move from one patch to another.

In the present paper, for modeling the evacuation process of a large event hall, six different scenarios are considered, in which we have varied the doors availability, the size and the structure of the evacuated population, the option to evacuate with a friend or a family member and the choice for evacuating through the closest door. The proposed agent-based model can be easily adapted to other types of large buildings by adjusting the characteristics of the event halls from the interface. In the same manner, using the interface of the model, one can easily define the type of population expected to attend such an event, by selecting different characteristics in terms of age, speed, rules of movement, or in terms of behavior, e.g. evacuating with another person or selecting or not the closest exit door as a result of the familiarity with the environment/panic, etc. In order to facilitate the shortest route to the exits, the paper proposes, based on the scientific literature, an "adapted cone exit" approach which provides to the agents the best route from any location they might be within the considered event hall.

The contribution of the paper is twofold. First, the paper shows that by using the agent-based approach the evacuation process from a large even hall can be easily simulated and observed by any interested party - thanks to the graphical interface offered by such a model, in which the elements involved in such a process are clearly identified (e.g. walls, exit doors, position of each evacuating agent, path chosen by each evacuated agent, obstacles, etc.). Second, the paper proposes an "adapted cone exit" approach through which the agents are able to find the best route from any point to the evacuation exit, making the simulation of the evacuating agents movement closer to a real-life situation. Compared to traditional evacuation methods, the advantages of the proposed approach used in the paper resides from the fact that: (1) the method clearly shows the differences in evacuation times depending on the number of exit routes available during the process; (2) the speed of movement of each individual can be customized according to the age category; (3) the simulated population takes into account the characteristics of a population under evacuation, being able to be diversified in terms of physical peculiarities (e.g. presence of people with locomotion impairment, keeping a specific distance among agents, evacuating with another agent, choosing or not the closest exit). The study has been conducted based on the court of events available in the cold seasons at the Roman Arenas,²

¹https://www.euronews.com/2020/10/30/colectiv-fire-romania-s-deadlynightclub-blaze-is-still-an-open-wound-five-years-on

²https://communistism.wordpress.com/2013/02/13/the-roman-arenasbucharest-romania/

Bucharest, Romania, one of the locations highly used for public events in Bucharest in both the warm and cold seasons.

A series of scenarios are used in the paper in order to determine the changes in the evacuation times due to various factors. Based on the simulation results, coupled with a visual analysis of the simulated scenarios, the identification of the main factors affecting the evacuation process can be easily made and the potential measures for improving the safety during the evacuation process can be timely observed.

The paper is structured as follows: section II provides a literature review related to the state of the art in evacuation. Section III briefly presents the selected location and the characteristics of the persons attending the events in the selected location. Section IV discusses the agent-based model assumptions, type of agents and implementation. Section V presents the scenarios considered for the simulations, while Section VI discuss the simulation results. The paper ends with limitations of the research presented in Section VII and discussions, conclusions and further developments in Section VIII.

II. THEORETICAL BACKGROUND AND STATE OF THE ART ON EVACUATION

Empirical data has shown that the main cause of death and injury of the participants in emergency evacuations is not the disaster itself, but the irrational and impulsive behavior of crowds under the influence of panic induced by the situation [12], [13]. In order to minimize the number of casualties, it is important that architects and engineers design the buildings in an optimal way to evacuate in panic conditions as safely as possible [14]. The behavior of certain individuals under stress is very difficult to predict since each person reacts differently depending on the environmental conditions. This is due to differences in age, gender, culture, physical and mental condition and background [15], [16]. Despite this, during an emergency, crowd behavior tends to follow some common characteristics independent of specific cases [17]-[21]. The emergence of a certain behavior has been observed in evacuation situations with crowded populations, where people tend to collide with each other on their way towards the exits, with the aim of faster escaping the location they are evacuating from in order to protect themselves [12], [13]. This makes an emergency evacuation even more dangerous than a coordinated one, significantly increasing evacuation times [11], [13], [14], [22].

Traditionally, crowd management and building evacuation are analyzed by observing the pedestrians moving in controlled spaces. The movement of these pedestrians is recorded and subsequently analyzed to produce mathematical analytical models that explain the behavior of crowds [19], [23]. These analytical models provide a better understanding for engineers and architects, helping them make decisions about the design and construction of buildings and evacuation procedures. However, the models in question are limited by the complexity of the buildings. The increased processing power of modern computers makes it possible to study the Two types of evacuation simulation models can be encountered in the scientific research: macroscopic and microscopic [24]. In the macroscopic models the crowds are considered as a single homogenous population, whereas the microscopic models rely on considering individual behavior and interactions.

From the macroscopic models, the social-force model, based on a molecular dynamics-based approach, is one of the most-known [25]-[27]. The model considers each pedestrian to be an unstructured particle whose motion is governed by Newton's equations [28]. A series of studies have been conducted to inspect individual-level interactions between people in a complex system with the purpose of exploring the mechanisms involved in the behavior of large human populations caught in an evacuation process [29]-[31]. The emergency evacuation simulations based on the social-force model have been widely used in scientific research [32]-[34]. One of the disadvantages of using such an approach is related to the fact that it does not properly incorporate the different individuals' behavior. Another one is related to the difficulty in implementation due to the relatively high number of nonlinear differential equations and the hypothesis needed for properly establishing those equations [24].

On the other hand, the microscopic models succeed in passing these shortcomings, but have been proven to become expensive due to the complexity they exhibit for very large populations or environments [24]. In order to reduce the complexity of the microscopic models, one can either choose to reduce the complexity of the characteristics and interactions of the considered population, or choose to reduce the complexity of the environment. Regarding the advantages of the microscopic models, one can name the emerging and sometimes unexpected behavior of the mass, which could have not been observed by simply reading the behavior rules of the individuals [35], [36]. Two types of microscopic models can be encountered: discrete – based on cellular automata [37]–[39] and continuous – based on agent-based modeling [40]–[42].

As for the studies featuring large-scale crowd evacuation using microscopic models, it can be observed that both cellular automata [43], [44] and agent-based modeling [45], [46] have been used in the scientific literature. A selection of the studies in each of the two modeling categories is discussed in the following.

Dang *et al.* [43] use cellular automata and virtual reality for simulating an evacuation scenario from a shopping mall. Based on the results, the authors state that the proposed approach is more appropriate to the type of selected environment than a simulation conducted using the Pathfinder software as it can easily incorporate a variable related to evacuees' environmental familiarity. One of the criticisms that can be brought to the approach is related to the movement of the evacuees, which can only be made into steps equal to one size unit. The start point, the endpoint and all the intermediate

points of the path are set to the center coordinates of the cells, the evacuees not being able to stop in any other point of the cell, as expected in a real-life situation. This limitation is common, in fact, to all the cellular automata models. As a result, the authors introduce in their paper the calculus of some interpolation points, which have the property of smoothing the path of the evacuees, but, as the authors state, the calculation accuracy of the chain navigation grid needs to be further optimized. Abdelghany et al. [44] incorporate genetic algorithms into a cellular automata model in order to find the optimal evacuation plan for a given situation. The proposed framework is applied to a hypothetical pedestrian facility with ten exits. The authors state that the approach can be used in practice, as it provides a superior evacuation plan when compared to a traditional approach [44]. Considering the location to be evacuated, one can state that the design of the location is rather simple, with no elements related to dynamics of the environment conditions, while the presence of obstacles reflects only partially a real-life situation. Wu et al. [47] extend a classical cellular automata model by incorporating a route choice model in the case of a large indoor space. The authors state that the proposed model has the ability to successfully incorporate various aspects related to evacuation. The limitations of the work are related to the interactions among the evacuating agents, which are not included. Considering a stadium scene, Zhou et al. [48] included in their model different types of emotions that can be experienced by the persons involved in an evacuation process, showing that during the evacuation, the individual emotions are changing, reaching calm for all the participants at the end of the simulation. Even though the introduction of emotions into an evacuation model makes it closer to a reallife situation, it is not clear to which extent one can succeed in knowing the personality of the individuals involved in such a process.

In the area of agent-based modeling, Ronchi et al. [45] use the Pathfinder software for modeling a large-scale evacuation from a music festival. The authors state that the modeling approach has been appropriate for the situation under investigation, mentioning that, in general, the evacuation models have limited capabilities in representing the complex behavior of the evacuees, as most of them do not account for propagation of information or social influence. Considering the model, more experimental data is needed to be introduced in the model for improving the reliability of the model results, as, in the current form, the model is only based on scarce literature. Ren et al. [49] propose an agent-based model created in Repast software. Based on the simulations, the authors observed a "faster is slower" phenomenon during the evacuation process. As the authors mentioned, the computational costs of such a model should be evaluated if one needs to simulate a more accurate model with an increased number of agents. The evacuation from elevated lecture halls has been simulated using an agent-based model in NetLogo by Delcea et al. [50] with the purpose of determining an adequate seat arrangement for a faster evacuation process.

The authors show that the proper combination between the seat arrangement and the position of the evacuation doors can diminish the evacuation time.

Siyam *et al.* [51] provide a comprehensive overview on agent-based simulation in the case of pedestrian evacuation, while Sharbini *et al.* [52] and Bayram *et al.* [53] provide an extended review on evacuation planning and management. Chen *et al.* [54] discuss the approaches, models and tools used for pedestrian evacuation in indoor emergency situations.

III. PREREQUISITES

Details related to the selected location and types of participants to the events held in the analyzed hall are discussed in the following.

A. LOCATION

The selected event hall is in a central area of Bucharest, Romania and it is usually used for unreeling international music festivals and international concerts – Fig. 1.

The central part of the location provides during a large period of the year (October-May) a heated tent with a length of approximately 45 m and a width of approximately 25 m, being able to accommodate up to 2500 persons per event. This part of the location is the one analyzed in the current paper and we will refer to it as the "hall" or the "tent" in the remainder of the paper.



FIGURE 1. The Roman Arenas Location.

The tent disposes of four ways of access, two of them being used for entering the location in normal use situations and the other two are dedicated to the emergency situations that might occur.

Fig. 2 presents the architectural scheme of the tent in which we have marked the two exit doors with A and B, the emergency doors with C and D, the stage is represented in blue, the two storage spaces located in the upper-left and bottom-left sides of the view are marked with cyan, the four pillars near the stage are colored in light-blue, the two food stands marked in orange, the sound-operators stand is marked in magenta, the two token-stands are colored in orange-red, while the three merchant stands located in the right side of the view are marked in light-pink.

The dimensions of the tent, the entrances and the obstacles present inside it are proportional with the real ones. It should be noted that the front of the tent is delimited by a fence guarded by security guards and access to the stage is allowed only to the persons with special permission (artists, managers, and photographers).

B. POPULATION

The persons attending the events held in the tent can be divided into two categories: "standard occupant" and "people with locomotion impairments" [45]. In the first category, one can include children, adults and seniors, while in the second category all the persons with locomotion impairment (wheelchair persons, persons with crutches/canes) are considered.

Based on the data reported by the firms organizing events in this location, it has been observed that most of the time, the average number of tickets sold for attending such an event is 2000 tickets. The distribution of the population differs according to the type of event held in the tent. An average distribution determined based on the events held in the prepandemic period has revealed that most of the participants belong to the adults category (approximatively 81.8%), followed by seniors (15.0%), children (3.0%) and people with locomotion impairments (0.2%). Besides the participants, an approximate number of 39 persons – part of the staff – are present, on average at the location.

IV. THE AGENT-BASED MODEL

An agent-based model is part of a class of computational models created to simulate the actions and interactions of autonomous agents (both individuals and collective entities) that aim to ascertain the effects on systems or other entities [55]-[57]. These models combine elements of game theory, complex systems, emergencies, computational sociology, operations research, and evolutionary programming [58]-[60]. Different methods are used to introduce randomness when stating the agents' behavior. In the recent literature it has been observed that agent-based models are used in non-computational scientific fields, including biology, ecology and social sciences [61]-[63]. Agent-based models are related to the concept of multi-agent systems or multi-agent simulations, as their purpose is to provide an explanation for the collective behavior of agents that follow simple rules.

Individual agents are characterized as being rational, namely their behavior is oriented for the good of their own interest, using simple decision-making or heuristic rules [64], [65].

Agent-based modeling, in some respects, is the most complex method through which a real environment is reproduced in a controlled environment. This type of modeling integrates environmental data with the behavioral and demographic aspects of the population to provide important data for theoretical studies and estimates. As a result, the agent-based models are virtual models that aim to reproduce the behavior of individuals in a given environment. They are more intuitive than mathematical or statistical models as they can represent objects in a similar manner in which they are seen in reality. Due to the emphasis in the last 30 years on object-oriented programming, agentbased models have become easier to implement, while their structure is easy to model based on mathematical and behavioral statistical models.

In the case of large event halls evacuation, an agent-based model has been created using NetLogo 6.2.2 The model's graphical user interface (GUI) is presented in Fig. 3, while a close-up on the elements in GUI are depicted in Appendix B. A series of elements can be configured from the interface and/or by uploading a text file containing the structure of the hall to be represented. Appendix A provides a table of nomenclature for the variables included as inputs and outputs in the agent-based model. The variables that refer to the characteristics of the evacuation population have been extracted from the scientific literature (e.g. speed), the variables needed for building the agent-based model's environment have been taken from the measurements of the concert hall, the variables related to population structure have been extracted from the statistics associated with the usual attendance for the events organized in the selected concert hall, while the variables related to the population's rules of movement have been taken from the evacuation simulations.

The configuration of the hall allows the user to establish the structural elements of the hall, such as the dimensions of the hall (length and width), the different stands, the position of the stage, the position of the doors and the availability of the doors.

In terms of population, one can configure the number of persons attending an event, the structure of the population attending the event (children, adults, seniors and persons with locomotion impairment), the presence/absence of staff persons, the location of the persons acting as staff at the event, the possibility to set-up families who might/might not evacuate together.

A series of assumptions have been stated in order to build the agent-based model, as presented in sub-section A. In terms of types of agents, two types have been used: *turtles* – agents possessing human characteristics in terms of evaluating the surrounding world and making needed evacuation decisions and *patches* – small pieces of ground through which the event hall has been divided into squares, having a series of characteristics which help the turtle agents to better understand the environment and to guide them to the available exits. The characteristics of each type of agent are described in sub-section B, along with the agents' movement rules.

A. ASSUMPTIONS

For building the agent-based model, the considered environment has been divided into small pieces of ground, having a square surface of $0.5m \times 0.5m$ [66], [67]. As a result of this division of the surface into small areas, the rest of



FIGURE 2. The Roman Arenas Tent Hall.



FIGURE 3. The Graphical User Interface of the Agent-Based Model.

the elements represented in the agent-based model (objects, doors, stands, stage, persons, etc.) are scaled to match multipliers of these values.

In particular, a person is represented in space by assuming an agent shoulder equal to 0.4558m, an assumption that is in line with previous studies involving human movement and behavior [45], [68]. As two persons cannot occupy the same space, even in the agent-based model, once an agent is located in an area given by the size of its shoulders, the space occupied by the agent cannot be, at the same time, occupied by another agent.

The walking area for the evacuees is represented by the flat floor of the tent, all the other objects installed by the organizers such as stands and stage cannot be overpassed by the agents in the evacuation process, representing obstacles that should be bypassed.

TABLE 1. Average evacuation time based on the category of participants.

Category	Average movement speed (m/s)	Speed range (m/s)
Child	0.90	0.90 ± 0.30
Adult	1.25	1.25 ± 0.30
Senior	0.80	0.80 ± 0.30
Persons with locomotion	0.79	0.79 ± 0.32
impairment		

The movement speed of the agents representing the "standard occupants" involved in an evacuation process has been taken from the data provided by Korhonen and Hostikka [68] as presented in Table 1. As not all the persons involved in an evacuation process have the same evacuation speed, we have considered the uniform probability distribution for the speed as proposed by Korhonen and Hostikka [68] – Table 1. This assumption is in line with the observation stated by Ronchi *et al.* [45] who mentioned that in order to account for the variability of the people abilities, the unimpeded walking speeds can be determined through the use of distributions. As for the persons with locomotion impairment, the average movement and the speed range have been taken from the research conducted by Hashemi [69]. Regarding the staff member, they have been associated with young adult persons, therefore, they are included in the adult population, having similar movement speed rules.

The agents moving towards the exits in the evacuation process are considered rational. As a result of this assumption, when the evacuation process starts, each agent starts moving towards the closest possible exit, using the shortest path to the chosen exit. In reality, the proper choice for the closest door can be made by the agents as a result of the experience/familiarity they have with the location as they might have attended other events in the same location or due to the implementation of some guidance systems by the organizers of the event – e.g., guiding light on the floor or a smartphone application.

In the case in which one or more of the exits are not available for evacuation, the agents will be aware of this situation and will proceed to the closest exit that can be used in the evacuation process. Also, the doors can be partially available for the evacuation, representing the case in which one half of the door is open, while the other half is closed due to various reasons, such as the impossibility to be opened as a result of a technical issue. For simulating such a situation, the user can choose from the interface, for each of the four exits, the "fully-opened," "half-opened" or "closed" option by using the chooser for each of the doors (*exit-A, exit-B, exit-C* and *exit-D*).

As the individual evaluation related to the closest exit position can be sometimes subjective when no support is offered in this process and the persons involved in an evacuation process cannot evaluate correctly all the time the closest exit - this might happen due to panic or stress generated by being in such a situation or due to other conditions generated by the emergency situation itself, such as the view of the exits is restricted by smoke and the evacuees are not familiar with the environment as they have not been previously attended another event in the tent - there might be cases in which the evacuees, even though they are assumed to be rational, might wrongly evaluate the closest evacuation door and choose another door which is not the closest door to their actual position. For such situations, the model can be adjusted from the interface and, for a given number of participants, the choice of the closest exit will be suboptimal, namely the randomly selected participant will not choose the closest exit, but rather one of the remaining exits.

The two situations, in which all the agents choose for sure the closest door and in which they choose the evacuation door following a probability distribution, are implemented in the agent-based model – through the *%-participants-choosingthe-closest-exit* slider – and will be used in simulations for better observing the differences in evacuation times.

When children or persons with locomotion impairment are present at the events held in the tent, it is assumed that they attend the event with an adult agent. As a result, when the population is randomly positioned in the environment, the children and persons with locomotion impairment are always located near an adult agent. In the case of an emergency two situations are possible: the adult might choose to help the child or the person with locomotion impairment to evacuate - case in which the two persons evacuate together, with the adult agent walking all the time behind the children or the person with locomotion impairment at a speed equal to the average speed between the two agents - or the adult choose not to help the other agents and each of the agents are moving towards the exits at their own speed. The choice for one of the two situations mentioned above can be established from the interface, by selecting "on"/"off" in the with-assist switcher.

For the situation in which the agents do not necessarily choose the closest exit, as the evacuation exit is determined using the probability distribution, and the adult agent decides to help the child agent or the person with locomotion impairment agent, the evacuation door will be the one selected by the adult agent for both agents.

As for the staff members, it is assumed that all the time they will choose the closest exit as they are familiar with the environment.

The evacuation process is considered complete when all the persons have evacuated using one of the available exit doors.

B. IMPLEMENTATION

The agent-based implementation in NetLogo 6.2.2 has been made through the use of the *patch* and *turtle* agents. The *patch* agents have been used for building-up the environment and for providing the information related to the exit doors to the *turtle* agents. In order to fulfill this purpose, the *patch* agents possess the characteristics in Table 2.

The shortest distance to the available exits is retained using the *exit-energy* variable of each patch through the use of a vector. The vector contains a number of values equal to the number of available exits. In order to better explain how the values are determined, we will consider in the following a hypothetical situation in which only one exit door is available for a certain room. As a result, the vector exit-energy contains only one value. In order to determine the shortest path (exitenergy) to an exit, an "adapted cone approach" based on Biner and Brun's "cone approach" [31] – Fig. 4 (A) – has been implemented, with a few differences.

TABLE 2. Patches characteristics in the agent-based model.

Name of	Range / Value	Short description
pcolor	black / dark blue / cyan / sky / magenta / lime / white / pink / orange / red	Patches in black are used to describe the floor of the event tent, while the ones in white for delimiting the tent, representing the walls. Two colors are possible for the exits: lime when the exit can be used for evacuation and red when the exit is closed. Different other colors can be used for the other types of objects, their color can be changed from the interface by using the input boxes.
object	floor / stage / storage / pillar / merchant stand / token stand / sound- operators stand / exit / wall	Indicates whether a patch is a part of the floor, exit or obstacles such as stage, storage, pillar, merchant stand, token stand, sound-operators stand or wall.
exit- index	0, 1, 2 or 3	Indicates the index of the exit. The variable is set only for the patches that represent the
		four exits in the model.
exit- energy	array of values between 0	Contains the shortest distance from the current patch to all the available exits. When the patch is initialized, the vector
	and 1000	contains the value 1000 for the distance to each available exit. During setup, the shortest distances are calculated using the "adapted cone exit" approach – as discussed in this section – and the vector is updated to the shortest distances to all the available exits.
plabel	value between 0 and 1000	Displays on each patch the value of the distance to the closest available exit. The value is 0 for the exits and 1000 for all the other patches representing the floor before the program starts computing the distances. The display of the <i>plabel</i> on each patch can be activated/deactivated from the <i>display-energy</i> switch located in the interface.

The first difference is related to changing the modality through which the numbers associated to the distances are being determined. In the approach we have used in the agentbased model, the available exits have received "0" for the patches representing them, while the numbers associated with

49366

the patches located near-by are increasing following a cone rule instead of decreasing as in [31] - Fig. 4 (B).

The second change has been due to watching the evacuation simulations made using the approaches (A) and (B) presented in Fig. 4. Based on simulations it has been observed that when multiple paths are available from an agent's current position to the closest available exit door, not all of them are equal in length. For example, in Fig. 5, an agent located in the upper-left corner of the image, near the obstacle marked in red, on the cell having the exit-energy of 17, can take two paths for reaching the exit patches marked with 0. Following the classical "cone exit" approach, any of the two paths (marked with yellow arrows) are possible to be followed as at every moment of time, the agent chooses among the adjacent patches, the patch with the smallest energy. But looking closer at the length of the paths, one can see that, in fact, they are not equal, as the upper path measures approximately 14 units (6 units horizontally, 5*1.4 units in diagonal and 1 unit vertically), while the lower path measures 17 units (6 units vertically and 11 units horizontally).

As a result, in the agent-based model we have implemented an "adapted cone exit" approach (Fig. 4 (C)) in which the exit-energy values are determined starting from the exit doors and using the smallest value between the exit-energy of adjacent patches updated with the distance between the current patch and each of the adjacent patches. A numerical exemplification is provided in Fig. 6. From the figure it can be observed that the exit-energy of the patch located in the left-bottom, marked with a darker shade of blue than the other patches, is determined based on the exit-energy of its adjacent patches for which the exit-energy has been determined in a previous step. As a result, the exit-energy for the abovementioned patch is the minimum between the exit-energy of the adjacent patches (1.4, 1 and 2) to which we have added the distance between the middle of selected patches (namely 1, 1.4 and 1). As a result, the *exit-energy* of the left-bottom patch is equal to 2.4.

Considering the same situation as in Fig. 5 (representing a part of the selected event hall from the agent-based model) and using the "adapted cone exit" approach, a partial view on the model developed in NetLogo 6.2.2 representing the lowerright exit (noted with exit D in Fig. 3) is presented in Fig. 7.

As it can be observed, an agent located in the upper-left cell of the picture will decide to follow the path marked by yellow arrows in order to arrive at the exit marked with 0. This choice is made by comparing the *exit-energy* of the adjacent cells, namely 13.1 and 14.5, and by choosing the smallest value (13.1 units) which belong to the path highlighted in Fig. 7.

The code for the proposed "adapted cone exit" is presented in Table 3.

In the case in which the agents are blocked and cannot advance anymore on the shortest path, in general due to the fact that they met on this path – for a short period of time – another agent that comes from ahead, the agents are able to choose the next shortest path available until they "escape" the blockage and then return to the shortest path to



FIGURE 4. Cone exit approach: (A) classical cone exit; (B) modified cone exit; (C) adapted cone exit.



FIGURE 5. Cone exit approach.



FIGURE 6. Determining the exit-energy using an adapted cone exit approach.

14.1>13.1>12.1>11.1>10.1>9.1>8.1	7.7	7.2	6.8	6.4	6	6	6	6	6.4
14.5	6.7	6.2	5.8	5.4	5	5	5	5	5.4
13.8	5.7	5.2	4.8	4.4	4	4	4	4	4.4
12.8	5.2	4.2	3.8	3.4	3	3	3	3	3.4
11.8	4.8	3.8	2.8	2.4	2	2	2	2	2.4
11.4 10.4 9.4 8.4 7.4 6.4 5.4	4.4	3.4	2.4	1.4	1	1	1	1	1.4
								~	

FIGURE 7. Adapted cone exit approach (patches view from the agent-based model built in NetLogo 6.2.2).

the selected exit door. This situation might appear in narrow areas such as the areas between the stands and the walls of the event hall.

TABLE 3. Adapted cone exit implementation.

```
Adapted cone exit implementation in NetLogo
to setup-energy
 ask patches with [object = "door"][
  compute-energy 0 self
 1
end
to compute-energy [energy-level floor-patch]
 (ifelse
  object = "door" [ set energy energy-level]
  object = "floor" [ set energy energy-level + distance myself]
  [])
 ask neighbors with [object = "floor" and energy > [energy] of myself
+ distance myself][
  compute-energy [energy] of myself self
 1
end
                        Ω
                                                           0
```



FIGURE 8. Adapted cone exit approach compared to chain navigation.

Compared to a chain navigation grid [43], the "adapted cone exit" offers a more realistic view of the participants evacuation process, with impact on the calculus of the overall evacuation time – Fig. 8.

The advancement of the agents to the exits is not conditioned by the position they have with respect to a certain patch (e.g., the agent does not have to be positioned in the middle of the patch at the beginning of the walk and does not have to arrive in the middle of a patch at the end of a time unit). As it will be discussed in the following, when the *turtle* agents are described, the agent's advancement within each moment of



FIGURE 9. Up-close look of the *turtle* agents.

time depend on its speed and is made by selecting each time the adjacent empty patch with the smallest *exit-energy*.

The *turtle* agents are used for representing the participants involved in an evacuation process. Each *turtle* agent is depicted using a circle of a different color in accordance with the type of participant to the event (child: green, adult: yellow, senior: grey, person with locomotion impairment: red, staff member: blue) – Fig. 9.

The *turtle* agents are randomly created within the event hall at the beginning of the simulation and the proportion of agents from each category is set from the *%-children*, *%-with-locomotion-impairment* and *%-seniors* sliders in the agent-based model interface. The staff members are created at random positions in the adjacent areas of their working locations.

For better shaping the *turtle* agents, a series of characteristics have been implemented in the agent-based model, as presented in Table 4.

After setting the environment and creating the agents to be evacuated, in a simulation run, the *turtle* agents proceed to the exit doors based on their *chosen-exit-index* and at their *speed*, being guided in the decision related to the chosen path by the values of the vector *exit-energy* stored at the level of every patch.

In order to test the choice for the shortest path to an exit, we have simulated the situation presented in Fig. 7 by manually placing an agent in the patch located in the upperleft corner of the figure. The moves of the agent towards the evacuation exit are provided in a few screenshots in Fig. 10 (it should be mentioned that we have not depicted the moves at every tick as the figure would have occupied more space).

From Fig. 10 it can be observed that the agent is following the same path as depicted in Fig. 7.

Depending on the speed of the agent, the time needed to cover the distance in Fig. 10 may vary.

As the turtle agents advance towards the selected exit at their own speed, the agents can arrive at the end of each moment of time (at each tick) anywhere inside a patch, not being conditioned to arrive in the middle of the patch.

49368

TABLE 4. Turtles characteristics in the agent-based mo	del.
--	------

Name of	Range /	Short description
variable	Value	
speed	[0, 1.24]	The speed range for the considered categories has been determined by translating the m/s measurement unit into patches/tick, considering the length of a patch of 0.5m and the duration of a tick of 2.5s [6]. As a result, the speed of a child is 0.72 ± 0.24 patches/tick, for a senior is 0.64 ± 0.24 patches/tick, for a senior is 0.64 ± 0.24 patches/tick, for a person with locomotion impairment 0.632 ± 0.256 patches/tick. When evacuating together, the speed of two agents is the average between their default speed. The maximum speed can be up to 1.24 patches / tick, representing 1.55 m/s when the turtle agent is an adult and has no other agent or obstacle in front of it. The minimum speed is 0 patches/tick and might occur when the agent is blocked and can no longer advance due to congestion or when it has evacuated. In the agent-based model, the uniform distribution of the speed has been introduced using the "random-float" function in NetLogo having as parameters the values of the speeds mentioned above. For example, in the case of a child agent, the NetLogo code for computing the speed is written as follows: set speed 0.72 + random-float 0.48 – 0.24 Random-float function generates a floating- point value between 0 and 0.48 and by subtracting 0.24, one obtains a floating-point value between -0.24 and 0.24. This value is afterwards added to the 0.72, resulting the
current-	[0, 1.24]	Indicates the speed of each agent at every tick.
speed category	adult, child, senior, locomoti on- impairm	Indicates the category of the agent.
travelled- distance	ent, staff ℝ ₊	Retains the distance travelled by the agent from its initial position to the current position. At the start of the simulation the travelled distance is zero for all the agents. At the end of the simulation, the travelled distance provides the distance travelled by the agent

The speed of an agent involved in an evacuation process is equal to its speed determined at the beginning of the simulation process but can be reduced at any time if in front

		from its initial position to the evacuation door.
chosen- exit-index	0, 1, 2 or 3	Retains the <i>exit-index</i> of the door to which the agent proceeds when involved in an evacuation process. The <i>exit-index</i> is a
		variable that can take 0, 1, 2 or 3 as values in the agent-based model, corresponding to the characteristics of the patches representing one of the four exits in the model.
my- teammate- ID	0,, number of agents in the model -1	In the case in which an agent is evacuating with another agent (e.g., the case in which the <i>with-assist</i> slider in the interface is "on" – the agents representing children or persons with locomotion impairment are evacuating with another agent) the variable retains the who
		number of the other agent. The <i>who</i> number is the unique identifier used by NetLogo for distinguishing among the turtle agents and starts from 0 to the number of agents created in the interface - 1.

TABLE 4. (Continued.) Turtles characteristics in the agent-based model.

of the agent there are agents which move at a lower speed on the same path and congestion appears.

The evacuation simulation stops when the last *turtle* agent has evacuated, no matter the door used for evacuation.

Regarding the evacuation process, two indicators can be determined through the agent-based model. The first one refers to the *overall evacuation time* and measures the time needed for the entire participants population to evacuate from the event tent, while the second one retains the *average evacuation time* and is determined by summing up all the individual evacuation times and dividing the result by the number of the agents. Both indicators are expressed in ticks in the agent-based model and can be transformed in seconds by dividing the result by 0.4.

For validating the model, a population composed by 53 adults (Fig. 11) have been evacuated from a small room, $4m \times 13m$, featuring a single exit door and a fixed obstacle placed in the central-left side of the room. An informed consent has been obtained from the participants.

The agent-based model has been adapted to fit the new environment and the agents, all adults, have been set on the random positions. A view of the evacuation process corresponding to one simulation run is presented in Fig. 12.

The overall evacuation time has been recorded by conducting three rounds of evacuation, with the evacuees placed at random locations within the considered space. The time has been compared with the one obtained through the use of the agent-based model adapted for the new situation, run for

	13.1	12.1	11.1	10.1	9.1	8.1	7.7	7.2	6.8	6.4	6		6	6	6.4
14.5							6.7	6.2	5.8	5.4	5	5	5	5	5.4
13.8							5.7	5.2	4.8	4.4	4	4	4	4	4.4
12.8							5.2	4.2	3.8	3.4	3	3	3	3	3.4
11.8							4,8	3,8	2,8	2,4	2	2	2	2	2,4
11.4	10.4	9.4	8.4	7.4	6.4	5.4	4.4	3.4	2.4	1.4	1	1	1	1	1.4
-				_							0	0	0	o	
14.1	13.1	12.1	11.	<u>_</u> 1	9.1	8.1	7.7	7.2	6.8	6.4	6	6	6	6	6.4
14.5							6.7	6.2	5,8	5,4	5	5	5	5	5.4
13.8							5.7	5.2	4.8	4.4	4	4	4	4	4.4
12.8							5.2	4.2	3.8	3.4	3	3	m	з	3.4
11.8							4.8	3.8	2.8	2,4	2	2	2	2	2.4
11.4	10.4	9,4	8.4	7.4	6.4	5.4	4,4	3.4	2.4	1.4	1	1	1	1	1.4
2											- 0	0	0	- 0	
14.1	13.1	12.1	11.1	10.1	9.1	8.)	7 .7	7.2	6.8	6.4	6	6	6	6	6.4
14.5							6.7	6.2	5.8	5.4	5	5	5	5	5.4
13.8							5.7	5.2	4.8	4.4	4	4	4	4	4.4
12.8							5.2	4.2	3.8	3.4	з	3	3	в	3.4
11.8							4.8	3.8	2.8	2.4	2		2	2	2.4
11.4	10.4	9.4	8,4	7.4	6.4	5.4	4,4	3.4	2.4	1.4	1	1	1	1	1.4
											0	0	0	0	
1.4.1	13.1	12.1	11.1	10.1	9.1	83	7.7	7.2	6.8	6.4	6	6	6	6	6.4
14.5							68	6.2	5.8	5.4	5	5	Ş	5	5.4
13.8							5.7	A	4.8	4.4	4	4	4	4	4_4
12.8							5.2	4.2	3.8	3.4	3	3	3	m	3.4
11.8							4.8	3.8	2.8	2.4	2			2	2.4
1,1.4	10.4	9.4	8.4	7.4	6.4	5.4	4.4	3.4	2.4	1.4	1	1	1	1	1.4
а 2											0	0	0	0	
14.1	13.1	12.1	11.1	10.1	9.1	8.3	7.7	7.2	6.8	6.4	6	6	6	6	6.4
14.5							68	6.2	5,8	5.4	5	5	5	5	5,4
13.8							5.7	25	4.8	4.4	4	4	4	4	4,4
12.8							5.2	4.2		3.4	3	n	3	3	3.4
11.8							4.6	3.8	2.8	2.4	2	2	2	2	2.4
11.4	10.4	9.4	8.4	7,4	6.4	5.4	4,4	3.4	2,4	1.4	1		1		1.4
											10	0	Q.	0	
14.1	13.1	12.1	11.1	10.1	9.1	8.1	7.7	7.2	6.8	6.4	6	6	6	6	6.4
14.5							617	6.2	5,8	5.4	5	5	5	5	5,4
13.8							5.7	32	4.8	4.4	4	4	4	4	4.4
12.8							5.2	4.2	38	3.4	3	3	3	3	3.4
11.8							4.8	3.8	2.8	254	2	2	2	2	2.4
11.4	10.4	9.4	8.4	7.4	6.4	5.4	4.4	3.4	2.4	1.4		1	1	1	1.4

FIGURE 10. Evacuation of a turtle agent using the adapted cone exit approach.



FIGURE 11. Evacuation of a small population made by adults from a small room.

10,000 times by using the BehaviourSpace option in NetLogo [56], [64]. In terms of time differences, it has been determined an average difference in evacuation time of 4.58%, which has been considered acceptable given the randomness in the speed of the agents and the position of the agents among the considered environment.



FIGURE 12. Agent-based simulation evacuation of a small population from a small room.

V. SCENARIOS

All the simulations have been conducted using the BehaviourSpace tool offered by NetLogo. Each scenario has been run 10,000 times and the average values have been reported in the paper, rounded to the nearest integer. The tool has been designed for conducting a large number of simulation experiments with the agent-based model [64].

An evacuation simulation can be considered complete when all participants in the event have been evacuated to a safe place. In general, a safe area is a location that is not affected by a disaster or emergency. In this article, the

49370

safe space is considered to be the outside of the event tent, which means that the simulation is completed once the entire population is evacuated from inside the tent.

Fig. 13 presents an evacuation simulation using the agent-based model for the considered event hall with 1500 participants and 39 staff members, all of them choosing the closest exit, in which the *display-energy* option has been set to "off" for providing a better view of the agents' paths. Additionally, the *display-path* option has been set to "on" for better observing the paths considered by the agents involved in the evacuation process. The images in Fig. 13 depict the evacuation environment and the position of the agents before the evacuation started (A), at a random moment of time (B), when the participants are only evacuating using Exit D (C) and at the end of the simulation process (D).

The scenarios considered in the study have been set up by varying different aspects related to the considered population, such as the doors availability (Scenario I), the size of the evacuated population (Scenario II), the structure of the population (Scenario III) and the choices made regarding evacuating with a friend/family member (Scenario IV) or not choosing the closest door (Scenario V). Each scenario has been divided into sub-scenarios for better highlighting how the variation in the selected indicators will impact the overall result. Each sub-scenario has been simulated 10,000 times and the average results have been reported in the paper by rounding them to the nearest integer. Additionally, a combined scenario has been set up (Scenario VI) in which some of the elements analyzed individually in Scenario I - Scenario V have been included in the same experiment. In all the scenarios, besides the overall evacuation time and the average evacuation time, the average distance travelled has been reported for giving more insight on the paths considered by the agents in their evacuation process. The results for the overall evacuation time and the average evacuation time have been transformed from ticks to seconds, while the values for the average distance travelled have been reported in meters instead of patches for ensuring a proper connection to the units used in our every-day life.

VI. SIMULATIONS RESULTS

In the following, the results obtained in the case of each scenario set-up in the previous section are presented and discussed.

A. SIMULATIONS RESULTS FOR SCENARIO I

For setting up the conditions for Scenario I, the population has been held equal to 2000 participants (81.8% adults, 15.0%seniors, 3% children and 0.2% persons with locomotion impairment – as it is in most of the events organized in the selected location) and 39 staff members. Additionally, it has been assumed that the children and the persons with locomotion impairment evacuate individually, all the agents choosing for sure the closest evacuation door.

The availability of the four doors has been varied within the sub-scenarios as presented in Table 5.



D) T = 420s

FIGURE 13. Evacuation simulation using the agent-based model for the event hall at different moments of time: A) 0s, B) 28s, C) 335s and D) 420s.

The baseline scenario, S-I.1., has considered the case in which all the exits are available for evacuation, and they can be completely used in the evacuation process.

TABLE 5. Exit doors availability.

	-					
Exit*						
А	В	С	D			
\checkmark	\checkmark	\checkmark	\checkmark			
×	\checkmark	\checkmark	\checkmark			
\checkmark	×	\checkmark	\checkmark			
\checkmark	\checkmark	×	\checkmark			
\checkmark	\checkmark	\checkmark	×			
×	×	\checkmark	\checkmark			
×	\checkmark	×	\checkmark			
×	\checkmark	\checkmark	×			
\checkmark	×	×	\checkmark			
\checkmark	×	\checkmark	×			
\checkmark	\checkmark	×	×			
	A	A B V V X V V X V V V X V X V X V X	ABC \checkmark \checkmark \checkmark \times \checkmark \times \checkmark \checkmark \times \checkmark			

 $*_{\checkmark}$ = available (open), \times = unavailable (closed)

Scenarios S-I.2. – S-1.5. assume that only three of the evacuation exits can be used at their full capacity, while in the S-1.6. – S-1.11. scenarios only half of the exits are available for the evacuation process. As the case in which only one door is available is hard to be encountered in practice, we have excluded this assumption when building the scenarios.

TABLE 6. Simulation results for scenario I.

		Indicator	
Scenario	Overall	Average	Average distance
	time	time	travelled
	(seconds)	(seconds)	(meters)
S-I.1. (Baseline)	547	192	26
S-I.2.	597	263	33
S-I.3.	882	294	35
S-I.4.	779	253	32
S-I.5.	694	245	32
S-I.6.	1096	487	50
S-I.7.	958	382	42
S-I.8.	857	356	41
S-I.9.	1038	397	43
S-I.10.	984	403	43
S-I.11.	641	275	38

As presented in Table 6, the best values for the reported indicators have been obtained for the S-I.1. situation in which all the four exit doors are available for the evacuation process. This result was expected, and it confirms the observations made in the research literature related to the fact that a higher number of exits will produce a faster evacuation process. The *overall evacuation time* is approximately 9 minutes and 7 seconds for S-I.1. On average, an agent needs 3 minutes and 12 seconds to evacuate, walking approximately 26m.

For the scenarios featuring the unavailability of one of the exits during the evacuation process, S-I.2. – S-I.5., it can be observed that the most unfavorable situation is the one in which Exit B is closed, which increases the *overall evacuation time* with 5 minutes and 35 seconds, representing an increasement of 61.24%. As expected, the values recorded for the *average evacuation time and average distance travelled* are higher in S-I.3. compared to S-I.1. with 1 minute and 42 seconds, respectively with 9m. The lowest impact on the *overall evacuation time* for S-I.2. – S-I.5 is recorded in the case of S-I.2., when Exit A is closed – lower with 4 minutes and 45 seconds than in the case of S-I.3 and higher with 50 seconds than in the case of S-I.1. Even though Exit A is larger than Exit C and Exit D, the impact of not being able to use Exit A for evacuation is smaller on the *overall evacuation time* than in the case of the other two exits (C and D), with up to 3 minutes and 2 seconds.

As the agent-based model offers a GUI that facilitates the observation of the agents' behavior during the entire evacuation process, a visual comparison between the S-I.2. – S-I.5. scenarios have been conducted on the purpose of identifying the differences in the recorded values of the three indicators reported in Table 6. Based on the observations, it has been determined that the prolonged evacuation time for some of the S-I.3. – S-I.5. scenarios compared with S-I.2. is due to the congestion around one of the available exit doors.

Specifically, it has been observed that in the case of S-I.2. the last agent evacuates through Exit D around T = 525s (with T = the moment of time), through Exit B at approximatively T = 580s and through Exit C at T = 597s. As for the S-I.3., it has been observed that no agent evacuates through Exit C after T = 432s, through Exit A after T = 553s, all the remaining agents evacuate through Exit D until T = 882s.

A similar situation is observed for S-I.4., where the agents are evacuating through Exits A and B until T = 348s, respectively T = 335s, while on Exit D until T = 779s. Lastly, for S-I.5., the agents evacuate through Exit A until T = 248s, Exit B until T = 513s and Exit C until T = 694s. Fig. 14 depicts the evacuation state exits in the considered scenarios S-I.2. – S-I.5 for the moment of time in which the agents are only left near one of the three evacuation exits.

Nevertheless, the evacuation doors position, and their width have a decisive role in the recorded overall evacuation time. For example, in the S-I.2. and S-I.5. scenarios, when the unavailable doors are Exit A and Exit D, the overall impact of these doors unavailability on the overall evacuation time is reduced compared to the S-I.3. and S-I.4. scenarios, when Exit B and Exit C are unavailable. These situations occur as, in S-I.2. Exit A, even though a large door, is located in the very front of the hall and does not "gather" as much agents as Exit B, a large door, which is opened in this scenario and is able to balance the absence of Exit A. A similar situation occurs in the case of S-I.5., where Exit D, even though a small exit placed in the middle-back of the hall with the potential of being the closest exit to more agents than Exit C, is closed and a series of agents decide to use Exit B, a large door which can balance the situation, providing enough space for agents evacuation. On the other hand, considering S-I.3. and S-I.4.



C) S-I.4. T = 348s



D) S-I.5. T = 513s

FIGURE 14. Evacuation simulations state and time for S-I.2. – S-I.5 when the agents are near one of the three evacuation exits: A) S-I.2., B) S-I.3., C) S-I.4. and D) S-I.5.

and the state of the simulations in Fig. 14, it can be observed that after T = 553s, respectively T = 348s, the only door with congestion is Exit D, which, due to its small dimensions,

does not succeed to produce a faster evacuation process. As a result, for this particular case of the considered event hall, one can consider enlarging Exit D as this will conduct to a smaller *overall evacuation time* in the case in which S-I.3. and S-I.4. scenarios occur – the scenarios with the higher *overall evacuation time*.

In terms of *average evacuation time*, it can be observed that the difference between the four situations (S-I.2. – S-I.5.) is up to 49 seconds, while the *average distance travelled* is up to 3m.

As for the scenarios in which two of the exit doors are unavailable, S-I.6. - S-I.11., it has been observed that the most unfavorable situations are the ones in which Exit B is closed, either in combination with Exit A (S-I.6.), Exit C (S-I.9.) or Exit D (S-I.10.). The overall evacuation time for the three situations in which Exit B is closed along with any of the remaining exit doors, range between 16 minutes and 24 seconds and 18 minutes and 16 seconds, being up to 24.26% higher than the case in which only Exit B is closed (S-I.3.) – Table 6. Considering the position of Exit B within the event hall and the size of this exit, the results were expected. Even in terms of average evacuation time and average distance travelled, the three situations (S-I.6, S-I.9. and S-I.10.) score the higher values among the scenarios featuring two closed exits, the average evacuation time being with up to 85.17% higher than the one recorded for S-I.3. where only Exit B was closed, while the average distance travelled is with up to 51.52% higher than in S-I.3.

The scenarios featuring two unavailable exit doors that produce the lowest overall evacuation time are S-I.11. and S-I.8., both having Exit D closed in combination with either Exit C or Exit A. While for the results obtained for S-I.8. (overall evacuation time of 857s, average evacuation time of 356s and average travelled distance of 41m) were expected given the fact that S-I.2. (having only Exit A closed) and S-I.5. (having only Exit D closed) produce the lowest overall evacuation time (597s, respectively 694s) among the scenarios with one unavailable door, being also lower than the overall evacuation time obtained for S-I.8. (where both Exits A and D are closed), the results obtained for S-I.11. in terms of *overall evacuation time* are surprising. By "surprising" we do not refer to the fact that the overall evacuation time for S-I.11. is lower than all the other scenarios in which two doors are unavailable (S-I.6.-S-I.10) as this result was expected given the fact that both exit doors C and D are smaller than the remaining opened exit doors A and B, which could ensure a faster evacuation process than the case in which exit doors C and D are opened, but surprising in comparison with S-I.5. scenario in which only Exit D is opened. For a more in-depth analysis, we have run several times the two scenarios, namely S.I.11. with an overall evacuation time of 641s, Exits D and C closed, and S-I.5. with an overall evacuation time of 694s, only Exit D closed. An evacuation state at the end of two simulations for the two scenarios is provided in Fig. 15.

As it can be observed in Fig. 15., the difference in the overall evacuation time in favor for the case in which only one door is closed instead of two derives from the imposed rule that the agents evacuate using the shortest path to the available exit. In the case of S-I.5. a series of agents decide to evacuate using Exit C as it is located near them, but it is a smaller exit, which leads to congestion, ignoring the fact that Exits A and B, situated at a longer distance than Exit C, are larger and might represent a better option. This choice conducts to an increased overall evacuation time. Fig. 16 depicts the evacuation time versus the number of evacuated participants for the two scenarios, considering time intervals of 25s. In the case of S-I.5. approximately 1193 participants evacuate in the first 250s, compared to 947 participants in the case of S-I.11., while between 250s and 600s the number of evacuees changes in favor of the S-I.11. (1076 participants vs. 773 participants in S-I.5.). On the other hand, at personal level, the average evacuation time and the average distance travelled are lower in the S-I.5. scenario (245 seconds, 32m) compared to S-I.11. (275 seconds, 38m). Considering these results, it can be stated that for a large number of participants, the evacuation process has been shorter when only one door is unavailable (S-I.5.), but, on the same time there have been some participants, a relatively small number, for which the evacuation process has been longer, increasing their personal risk. Based on these observations, if needed, the agent-based model can be adjusted in order to incorporate congestion as a second criterion when choosing the evacuation door. This improvement applies to some evacuation cases in which the participants have the ability to observe the congestion from the other doors and change their mind related to the selected evacuation door. In reality, in large crowd evacuation cases, due to the large number of participants, it is hard to be in a congestion situation and to have the visibility to other evacuation doors. Even more, for the cases in which the emergency involves a lot of smoke or the lack of light or the occurrence of other hazardous conditions, such as fallen large objects or ceiling, the adjustment of the selected door based on observing the congestion around the other evacuation doors does not seem a realistic scenario.

As a general result based on the data provided in Table 6, it can be stated that the position, the number and the width of the evacuation exits impact the evacuation time, but as observed in the results of the simulations made for Scenario I, the results are highly dependent on the overall structure of the event hall, which implies the need for building a model that reproduces as precise as possible the characteristics of the simulated environment.

B. SIMULATIONS RESULTS FOR SCENARIO II

Scenario II discusses the differences in evacuation time when the number of participants varies. Table 7 presents the cases





B) S-I.11. T = 641s

FIGURE 15. Evacuation simulations state and overall evacuation time for S-I.5. and S-I.11: A) S-I.5., B) S-I.11.



FIGURE 16. Evacuation time versus the number of evacuated participants for S-I.5. and S-I.11.

considered. The S-II.3. situation is the baseline situation as it reproduces the general audience of the event hall, the results being the same as in the S-I.1. scenario. The other three scenarios considered feature the highest number of participants that can attend an event (2500 participants, S-II.4.) and a lower number of participants (1000, respectively 1500 participants). The structure of the participants is the same in all the considered scenarios: 81.8% adults, 15.0% seniors, 3.0% children and 0.2% persons with locomotion

TABLE 7. Simulation results for scenario II.

Comorio	S-II					
Scenario	S-II.1.	S-II.2.	S-II.3. (Baseline)	S-II.4.		
Number of participants	1000	1500	2000	2500		

TABLE 8. Simulation results for scenario II.

		Indicator	
Scenario	Overall evacuation	Average evacuation	Average distance
	(seconds)	(seconds)	(meters)
S-II.1.	287	104	17
S-II.2.	418	148	22
S-II.3. (Baseline)	547	192	26
S-II.4.	675	237	29

impairment, while all the four evacuation doors are available during the entire evacuation process. A number of 39 staff members add to each considered scenario.

The results obtained after 10,000 simulations are presented in Table 8, rounded to the nearest integer. Compared to the baseline scenario, S-II.3., a 50% reduction of the number of participants, S-II.1., conducts to a reduction of the *overall evacuation time* of 47.53%, with a comparable reduction of the *average evacuation time* of 45.83% and a reduction of the *average travelled distance* of 34.62%.

On the other hand, a reduction of 25% of the number of participants compared to baseline, reduces the *overall evacu*ation time by 23.58%, the average evacuation time by 22.92% and the average travelled distance by 15.38% (S-II.3. vs S-II.2.), while an increasement of 25% increases the overall evacuation time by 23.40%, the average evacuation time by 23.44% and the average travelled distance by 11.54% (S-II.3. vs S-II.4.). Based on the recorded values, it can be stated that the changes in the number of evacuated persons impact all the three indicators, the less affected indicator being the average distance travelled.

As for the *overall evacuation time*, Fig. 17 presents the evolution of the number of evacuated participants over time in the four scenarios.

Based on Fig. 17, it can be observed that for a period of time ranging between 75s (S-II.1.) and 225s (S-II.4.), the number of evacuated persons per unit of time is 149 - 153 persons, decreasing until the end of the simulation. The period of time corresponds to the period in which all the four exits are used at their maximum capacity and can be easily identified by running the agent-based model and watching the evacuation



FIGURE 17. Evacuation time versus the number of evacuated participants for S-II.1. – S-II.4.

process. After this period of time, step-by-step, the agents choosing Exit A, B and C finish the evacuation. In all the situations, Exit D is the last door through which the participants are still evacuating. As a result, the evacuation time is prolonged due to the Exit D position and length, with a more accentuated impact on the situations in which the number of participants is large.

Even in the case of Scenario II it can be observed that the number of participants has an impact on the considered indicators, increasing their values as the number of participants increases. Besides this influence, the structure of the room plays an important role as the indicators might increase slowly if the position of the evacuation doors and their width are adjusted to better fit an evacuation situation.

C. SIMULATIONS RESULTS FOR SCENARIO III

As in the previous scenarios, we have held the same baseline scenario as in S-I.1. (called now S-III.1.) and we have built upon varying the elements related to the structure of the evacuation population, keeping them close to the possible real structures of the crowds of persons attending such large events. The sub-scenarios to be analyzed are presented in Table 9.

The simulations results are reported in Table 10. The first observation is that, as the number of adults decreases, the *overall evacuation time* increases. This change in the *overall evacuation time* can be due to the values of speeds set for the considered categories, with the adults being the category with the highest speed. A decrease of the adult population from 81.8% (S-III.1.) to 15.0% (S-III.6.) determines an increasement in the *overall evacuation time* of 15.54%. At personal level, the *average evacuation time* increases with 16.67%, while the *average distance travelled* is almost the same, the small changes reported in this indicator might be due

	Category							
Scenario	Adults	Seniors	Children	Persons with locomotion impairment				
S-III.1. (Baseline)	81.8%	15.0%	3.0%	0.2%				
S-III.2.	60.0%	30.0%	9.0%	1.0%				
S-III.3.	45.0%	45.0%	9.0%	1.0%				
S-III.4.	35.0%	35.0%	29.0%	1.0%				
S-III.5.	30.0%	60.0%	9.0%	1.0%				
S-III.6.	15.0%	81.0%	3.0%	1.0%				

TABLE 10. Simulation results for scenario III.

	Indicator			
Scenario	Overall evacuation time (seconds)	Average evacuation time (seconds)	Average distance travelled (meters)	
S-III.1. (Baseline)	547	192	26	
S-III.2.	571	200	26	
S-III.3.	592	207	25	
S-III.4.	596	210	25	
S-III.5.	611	215	25	
S-III.6.	632	224	25	

to the stochastic simulation and rounding to the closest integer.

As for the S-III.4. situation, in which a quasi-balanced population (between adults, seniors and children participants plus the persons with locomotion impairment) has been considered, it can be observed that the *overall evacuation time* increases by 8.96% compared to the baseline. The *average evacuation time* increases by 9.38%, while the *average distance travelled* has almost the same value as in S-III.1.

As the speed of the population evacuating determines the values of the considered indicators, it can be said, once more, that knowing and evaluating properly the characteristics of the evacuated population and adequately introducing them in the agent-based model can improve the accuracy of the simulation results.

D. SIMULATIONS RESULTS FOR SCENARIO IV

Scenario IV refers to the case in which the children and the persons with locomotion impairment are helped for evacuation by an adult agent. The children and the persons with locomotion impairment are created in the agent-based model near an adult – as they were even in the case of the previous scenarios – but, in this case, they evacuate together with the adult, choosing the same door as the adult with whom they are evacuating, with the adult behind the child or the locomotion impairment person. The speed of the agents evacuating



FIGURE 18. Overall evacuation time in S-III vs. S-IV sub-scenarios.

		Indicator			
Scenario	Overall evacuation time (seconds)	Average evacuation time (seconds)	Average distance travelled (meters)		
S-IV.1.	541	190	26		
S-IV.2.	550	193	26		
S-IV.3.	568	199	26		
S-IV.4.	530	187	26		
S-IV.5.	591	207	25		
S-IV.6.	624	221	25		

TABLE 11. Simulation results for scenario IV.

together is determined as the average speed between their bydefault speeds. Besides these assumptions, we have kept the same structure of the population as in Table 9, with the only change that we have named the sub-scenarios using "IV" instead of "III."

The results reported in Table 11 have been determined by running 10,000 each sub-scenario and determining the average value of the indicators rounded to the first integer value. A comparison between the *overall evacuation time* in scenarios III and IV can be observed in Fig. 18.

A first observation based on Fig. 18 is that the *overall* evacuation time is smaller in the case in which the children and the persons with locomotion impairment are evacuating with an adult agent. The improvement in the *overall evacuation time* is up to 1 minute and 6 seconds. The largest difference between the considered sub-scenarios is recorded between S-III.4. and S-IV.4., both of them featuring a situation in which the children plus the persons with locomotion impairment represent a consistent part of the evacuation population (30.0%). Smaller differences are recorded for the cases in which the children and the persons with locomotion



FIGURE 19. Evacuation time versus the number of evacuated participants for S-III.4. and S. IV.4.



FIGURE 20. Evacuation time versus the number of evacuated participants (adults, children, and persons with locomotion impairment) for S-III.4. and S. IV.4.

impairment persons represent a small percentage of the population (e.g. 3.2% in S.III.1. and S.IV.1.).



FIGURE 21. Evacuation time versus the number of evacuated adult participants for S-III.4. and S. IV.4.



FIGURE 22. Evacuation time versus the number of evacuated children and persons with locomotion impairment participants for S-III.4. and S. IV.4.

Studying more in depth the S-III.4. vs S-IV.4. evacuation process by watching step-by-step the agent-based simulations, it can be observed that an increased number of agents are evacuating in the first 250 seconds in S-IV.4. case than in the S-III.4. (based on simulations the number of evacuated agents is approximatively 1494 in S-IV.4. vs. 1330 agents in S-III.4.) - Fig. 19.

Considering the category of agents evacuating together (namely the adults, children and persons with locomotion impairment) and comparing the evacuation time versus the number of evacuated persons, it can be observed that the S-IV.4. is favorable compared to S-III.4. as in the first 200 seconds of the evacuation process 897 agents evacuate in S-IV.4. compared to 744 agents in S-III.4. – Fig. 20.

By dividing the category represented in Fig. 20 into adults (Fig. 21) and children and persons with locomotion impairment (Fig. 22), it can be observed that the situation in which the children and persons with locomotion



FIGURE 23. Evacuation time versus the number of evacuated senior participants for S-III.4. and S. IV.4.

impairment evacuate with adults is a beneficial one for this vulnerable category as in the first 175 seconds of evacuation a number of 358 children and persons with locomotion impairment evacuate in S-IV.4. compared to 251 persons in S-III.4.

It can be observed that even the adult category has benefits from the situation in which they evacuate with other agents as it has been observed that by helping the other agents, the entire evacuation process speeds up. As a result, after 200 seconds since the start of the evacuation process, in S-IV.4. a number of 494 adult agents have evacuated compared to 442 adult agents in S-III.4. – Fig.21.

Moreover, the senior category, even though it does not contribute directly to the new situation as they evacuate as single persons, has to gain from the fast evacuation process resulted from the fact that the children and persons with locomotion impairment evacuated with an adult agent – Fig. 23. As a result, after the first 200 seconds since the start of the evacuation, 434 senior agents have evacuated in S-IV.4. compared to 389 senior agents in S-III.4.

Based on the simulations results it can be concluded that when the children and locomotion impairment agents are evacuating with another adult agent the overall evacuation process is improved by minimizing the *overall evacuation time*. As for the *average evacuation time*, a 23 seconds improvement has been recorded in S-IV.4. compared to S-III.4., with smaller improvement times in the other considered cases. The average distance travelled remains almost the same and it is due to the rules regarding the random initial placement of the agents within the environment and to the fact that the agents choose all the time the closest evacuation door.

E. SIMULATIONS RESULTS FOR SCENARIO V

For Scenario V it has been assumed that not all the agents are choosing the closest evacuation door. The situation might

			S-V		
Scenario	S-V.1. (Baseli ne)	S-V.2.	S-V.3.	S-V.4.	S-V.5.
Number of participants choosing the closest door	100%	95%	90%	85%	80%

TABLE 12. Simulation results for scenario V.

TABLE 13. Simulation results for scenario V.

Indicator			
Overall	Average	Average	
evacuation	evacuation	travallad	
(seconds)	(seconds)	(meters)	
(seconds)	(seconds)	(incicits)	
547	192	26	
558	194	27	
567	197	28	
581	202	29	
613	209	30	
	Overall evacuation time (seconds) 547 558 567 581 613	IndicatorOverall evacuation time (seconds)Average evacuation time (seconds)547192547192558194567197581202613209	

arise when the participants to the event are not familiar with the environment or in the cases in which due to the low visibility, they are not able to localize the closest exit. Not selecting the closest door can be also a result of panic or a result of a decision to go to a specific area in which one expects to find a relative or a friend. As Haghani and Sarvi [70] mentioned in their study, even though in normal times the proximity to a destination is the most prominent factor when selecting an exit, it has been proven to be completely irrelevant in emergency decisions. The reasons for not choosing the closest exit can be numerous, and one might envision some more considering the psychological aspects involved by being in an emergency situation. We will not discuss them in this part, as the focus is on the changes in the evacuation time when this situation might arise.

The envisioned sub-scenarios are presented in Table 12. As in the previous cases, for the baseline case, S-V.1., we have considered the same conditions as is S-I.1., while for the S-V.2. - S-V.5. sub-scenarios we have reduced the percentage of the agents choosing the closest door.

As expected, not choosing the closest door leads to increasement in all the three indicators – Table 13.

Comparing S-V.1. with S-V.5., it can be observed that a reduction of 20% of the participants choosing the closest door conducts to an increasement in the *overall evacuation time* of 1 minute and 6 seconds (12.07%), increasing the average



FIGURE 24. Evacuation time versus the number of evacuated participants for S-V.1. and S-V.5.

TABLE 14. Simulation results for scenario VI compared to S-I.1.

	Indicator		
Scenario	Overall	Average	Average
	evacuation	evacuation	distance
	time	time	travelled
	(seconds)	(seconds)	(meters)
S-I.1. (Baseline)	547	192	26
S-VI	875	325	41

evacuation time with 17 seconds (8.85%) and the average distance travelled with 4 m (15.38%).

In terms of evacuated persons versus the evacuation time, when comparing S-V.1. with S-V.5. (Fig. 24), it can be observed that in the first 250 seconds since the evacuation started, the when the agents are selecting the closest door, a number of 1439 agents are evacuated, while when only 80% of them are choosing the closest door, a number of 1305 agents are evacuated (with approximatively 9.31% agents less).

Based on the simulation results for this scenario, it can be stated that as the agents do not choose the closest door, the *overall evacuation time*, *average evacuation time* and the *average distance travelled* increase. As a result, the interested parties in offering a safe evacuation process should consider applying the proper means through which during such an emergency the population is guided to the closest exit, which results in diminishing the evacuation time and life savings.

F. SIMULATIONS RESULTS FOR A COMBINED SCENARIO

A combined scenario has been set up in this section, in which elements from the individual scenarios have been considered in order to demonstrate once more that the cumulus of



FIGURE 25. Evacuation time versus the number of evacuated participants for S-I.1. and S-VI.

factors that might appear in an evacuation can conduct to a completely different result than the one considered in an "ideal" case in which all the things go right. As the purpose was not to set up values for the variables that are hard to be encountered in an emergency, a "mild" scenario has been considered. In this scenario, S-VI., we have started from the baseline scenario S-I.1., in which we have altered some of the assumptions. As a result, it has been assumed that one of the emergency exit doors is not available to be used in the evacuation process (Exit C), while one of the main exits can only be partially used for evacuation (Exit B), that the children and the locomotion impairment persons evacuate with an adult and that only 80% of them are choosing the closest door.

The results obtained are summarized in Table 14. It can be observed that for this scenario, S-VI, the *overall evacuation time* is 14 minutes and 35 seconds (with 59.96% higher than in the S-I.1.), while the *average evacuation time* is 5 minutes and 25 seconds (with 69.27% higher than in the S-I.1.), while an *average distance travelled* of 41m (with 57.69% higher than in S-I.1.).

As for the evolution of the evacuation process in S-I.1. compared to S-VI, Fig. 25 presents the number of evacuated agents versus time. It can be observed that in the S-VI scenario the number of agents evacuated in the first 400 seconds is 1401 agents, compared to 1848 agents in the S-I.1.

Once more, it can be stated that for ensuring a proper evacuation process, the knowledge related to the location to be evacuated is of utter importance, along with the characteristics and the behavior of the evacuated persons. As shown in the cases presented above, small changes in different indicators can conduct to various changes in the result, while the combination of the different categories of changes can produce totally different results than expected.

As a result, the proposed agent-based model can serve as a tool through which the different aspects related to the modeled event hall are represented in a simplified manner and the characteristics of the evacuees are better shaped. Through the increased number of simulations one can conduct in a short amount of time and through the visual interface offered by the agent-based model, one can better understand which are the elements to be considered and what can be done in order to improve the values of the considered indicators, with a direct result in improving the entire evacuation process.

The model has some limitations related to the elements considered in the evacuation process and can be further improved by including different states in which the agents can be as a result of the feelings they have during such an event, states that can determine a specific behavior for each agent.

VII. LIMITATIONS OF THE STUDY

A potential limitation of the study is related to the fact that the validation of the model has been made only on an adult population. This limitation has not been further investigated in the present paper as, through the validation made on the adult agents, it has been observed that the differences between the overall evacuation time in the case of the adapted agentbased model and the on-site simulations have been below 5%. As a result, it is expected to have the similar differences between the real-life simulations and the agent-based model in the case of the other categories of people considered in the model. Even more, the agents' speed is a parameter in the model, so anyone interested in conducting his/her own simulations on a population already known, can easily introduce the speeds of the considered categories of persons and observe the behavior of the evacuees and the values of the resulting variables.

Another limitation is given by the number and types of variables used in the agent-based model. We acknowledge that with the increasement of the number of variables considered in the study, the incidence of each variable can be better shaped. Even more, it would be interesting to observe the result of the cumulated factors on the evacuation time. Additional statistical analysis can be conducted on the obtained results in accordance with the specific of the study one decides to conduct.

Lastly, the paper focuses on a general evacuation process, as described above. Considering some specific evacuation processes (e.g. fire broken out in fixed points of the room), one can observe that the impact of the considered factors might change. This limitation refers to the results obtained for the output variables. As for the agent-based model, it can be adapted to better fit the situation one needs to simulate.

VIII. CONCLUDING REMARKS

The paper discusses the use of agent-based modeling and simulation in the context of large event hall evacuation for better understanding the overall evacuation process and for observing the elements that can contribute to a prolonged evacuation process.

On this purpose, an agent-based model is built in NetLogo 6.2.2 and an "adapted cone exit" approach is proposed, which is useful in guiding the agents to the closest exit. A series of scenarios have been envisioned and simulated for better observing how the changes in different aspects of the evacuation process can influence the *overall evacuation time*, the *average evacuation time* and the *average distance travelled* by the agents.

Specifically, for the selected event hall, it has been observed throughout the simulations that when the number of participants attending an event is increased, Exit D is the last door through which the evacuation takes place. As a result, it is recommended, that in the future, if possible, the width of this exit to be increased in order to reduce the evacuation time. Even more, as Exit B plays an important role in the evacuation process, it is advisable for the persons in charge to take all the measures for keeping it available for the emergency situations by checking the space around the door and making sure that the door is not blocked or accidentally locked.

As choosing the closest door by the evacuees has a positive impact on the evacuation time, it is advisable that the persons in charge should take all the measures to ensure that the attendees to the event are familiar with the positions of the evacuation doors – this can be made by placing a small map on the back of the ticket or by playing a small video on the screen just before the start of the event. Nevertheless, helping other persons to evacuate should be among the elements to be kept in mind for decreasing the evacuation time. As a result, increasing the awareness for helping other in-need persons can be made through short videos or messages posted in the surrounding area of the event.

Given the flexibility of the agent-based model, the behavioral aspects of the evacuees and the environment characteristics of the evacuated space can be effectively represented in the model. In particular, given the simulation results and the visual analysis that can be conducted individually on each scenario, the predominant factors affecting the evacuation process can be identified and the potential measures can be properly evaluated prior to be implemented in practice. As a result, the evacuation process safety levels can be improved.

Future research can expand the model by including more information related to the vulnerable population, the feelings and reactions one might experiment in an evacuation process with effect on the evacuation behavior or the impact of specific events (e.g., different levels of gas concentration, the occurrence of multiple fire points within the event hall) on people's behavior and evacuation process. The impact of each newly considered element can be further put in connection with the resulting variables in order to better observe which of these elements are the ones leading to the greatest changes in the resulting variables. For this type of analysis, grey systems theory can be used, by computing several degrees of grey incidence, pointing in this manner the hierarchy of the elements with the highest incidence on the evacuation time.

APPENDIX A Table of NOMENCLATURE

TABLE 15. Table of nomenclature.

Name of variable	Meaning		
average distance travelled	Average distance travelled by each		
	turtle agent		
average evacuation time	Average evacuation time for each		
	turtle agent		
category	Category of each turtle agent		
chosen-exit-index	Exit chosen by each turtle agent		
current-speed	Current speed of each turtle agent		
display-energy	Displays the distance from the current		
	patch to the closest exit		
display-path	Enables the draw of the path of each		
	turtle agent from its initial position to		
	its current position		
exit-energy	Energy of the exit		
exit-index	Index of the exit		
my-teammate-ID	Identifier of the turtle agent evacuating		
	with the current agent		
object	Type of patch		
overall evacuation time	Time needed for the entire population		
	to evacuate		
patches	Agents used for building the		
	environment		
pcolor	Color of each patch agent		
plabel	Label of the patch		
speed	Speed of each turtle agent		
travelled-distance	Distance travelled by each turtle agent		
turtles	Evacuating agents		
with-assist	Evacuation situation in which the		
	agents evacuate together with other		
	agents		
%-children	Percent of children in all the		
	evacuation population		
%-with-locomotion-	Percent of persons with locomotion		
impairment	impairment in the evacuation		
	population		
%-participants-choosing-the-	Percent of participants choosing the		
closest-exit	closest exit		
%-seniors	Percent of seniors in the evacuation		
	population		

APPENDIX B CLOSE-UP ON AGENT-BASED MODEL GRAPHICAL USER INTERFACE

Input

Select the colors for the scheme:				
floor-color	stage-color	storage-color	P	illar-color
0 (black)	105 (blue)	85 (cyan)		98 (sky + 3)
foh-color 125 (magenta)	fence-color 19 (red + 4)	exit-color 65 (lime)		
Exits:				
exit-A full-opened ▼	xit-B full-opened 🛛 🔻	exit-C full-opened	exit-D full-opened	d 🔻
Participants:				
no-of-participants 2000 partici	ipants %-participants	s-choosing-the-closest-exit	90 %	
%-children 24 %	%-with-locomotion-in	mpairment 1.9 %	%-seniors	13 %
Participant behaviour:				
On with-assist				
Structure of building input file:	x_min y_min width h	eight "name" color no_	_of_staff	
No. Adults No. Seniors	No. Children	No. Participants with Locomot	ion Impairment	
1222 260	480	38		
Decels are set				1
10 People evacuat	ed based on their cate	gones adult child senior impairm staff	No. Staff 39 No. Helpe 0	ers
0		10		
Duration Avg. durat 0 0	ion Avg. distance 0		_	

FIGURE 26. Close-up on graphical user interface – left side.



FIGURE 27. Close-up on graphical user interface – right side.

REFERENCES

- E. Urbina and B. Wolshon, "National review of hurricane evacuation plans and policies: A comparison and contrast of state practices," *Transp. Res. A, Policy Pract.*, vol. 37, no. 3, pp. 257–275, 2003, doi: 10.1016/S0965-8564(02)00015-0.
- [2] S. Nordbeck, Computer Cartography: Shortest Route Programs. Lund, Sweden: Royal Univ. of Lund, Department of Geography, 1969.
- [3] S. A. Parr and E. Kaisar, "Critical intersection signal optimization during urban evacuation utilizing dynamic programming," *J. Transp. Saf. Secur.*, vol. 3, no. 1, pp. 59–76, Mar. 2011, doi: 10.1080/19439962.2011.532297.
- [4] D. E. Knuth, "A generalization of Dijkstra's algorithm," *Inf. Process. Lett.*, vol. 6, no. 1, pp. 1–5, Feb. 1977, doi: 10.1016/0020-0190(77)90002-3.
- [5] M. Dorigo, "Ant colony optimization," *Scholarpedia*, vol. 2, no. 3, p. 1461, 2007, doi: 10.4249/scholarpedia.1461.
- [6] C. Delcea, L.-A. Cotfas, L. Craciun, and A. G. Molanescu, "An agent-based modeling approach to collaborative classrooms evacuation process," *Saf. Sci.*, vol. 121, pp. 414–429, Jan. 2020, doi: 10.1016/j.ssci.2019.09.026.
- [7] N. Li and R.-Y. Guo, "Human behavior during emergency evacuation: Cell transmission model," *IEEE Access*, vol. 9, pp. 42463–42482, 2021, doi: 10.1109/ACCESS.2021.3059326.
- [8] M. Isobe, T. Adachi, and T. Nagatani, "Experiment and simulation of pedestrian counter flow," *Phys. A, Stat. Mech. Appl.*, vol. 336, nos. 3–4, pp. 638–650, May 2004, doi: 10.1016/j.physa.2004.01.043.
- [9] L. Z. Yang, D. L. Zhao, J. Li, and T. Y. Fang, "Simulation of the kin behavior in building occupant evacuation based on cellular automaton," *Building Environ.*, vol. 40, no. 3, pp. 411–415, Mar. 2005, doi: 10.1016/j.buildenv.2004.08.005.
- [10] D. Helbing and P. Molnár, "Social force model for pedestrian dynamics," *Phys. Rev. E, Stat. Phys. Plasmas Fluids Relat. Interdiscip. Top.*, vol. 51, no. 5, pp. 4282–4286, May 1995, doi: 10.1103/PhysRevE.51.4282.
- [11] V. J. Blue and J. L. Adler, "Cellular automata microsimulation for modeling bi-directional pedestrian walkways," *Transp. Res. B, Methodol.*, vol. 35, no. 3, pp. 293–312, Mar. 2001, doi: 10.1016/S0191-2615(99)00052-1.
- [12] N. R. Johnson, "Panic at 'the who concert stampede': An empirical assessment," *Social Problems*, vol. 34, no. 4, pp. 362–373, Oct. 1987, doi: 10.1525/sp.1987.34.4.03a00040.
- [13] J. D. Sime, "Crowd psychology and engineering," Saf. Sci., vol. 21, no. 1, pp. 1–14, Nov. 1995, doi: 10.1016/0925-7535(96)81011-3.
- [14] D. Nilsson and A. Johansson, "Social influence during the initial phase of a fire evacuation—Analysis of evacuation experiments in a cinema theatre," *Fire Saf. J.*, vol. 44, no. 1, pp. 71–79, Jan. 2009, doi: 10.1016/j.firesaf.2008.03.008.
- [15] D. Helbing, I. Farkas, and T. Vicsek, "Simulating dynamical features of escape panic," *Nature*, vol. 407, no. 6803, pp. 487–490, Sep. 2000, doi: 10.1038/35035023.
- [16] E. D. Kuligowski, X. Zhao, R. Lovreglio, N. Xu, K. Yang, A. Westbury, D. Nilsson, and N. Brown, "Modeling evacuation decisions in the 2019 kincade fire in California," *Saf. Sci.*, vol. 146, Feb. 2022, Art. no. 105541, doi: 10.1016/j.ssci.2021.105541.
- [17] S. P. Hoogendoorn and P. H. Bovy, "Pedestrian route-choice and activity scheduling theory and models," *Transp. Res. B, Methodol.*, vol. 38, no. 2, pp. 169–190, Feb. 2004, doi: 10.1016/S0191-2615(03)00007-9.
- [18] F. Venuti and L. Bruno, "Crowd-structure interaction in lively footbridges under synchronous lateral excitation: A literature review," *Phys. Life Rev.*, vol. 6, no. 3, pp. 176–206, Sep. 2009, doi: 10.1016/j.plrev.2009.07.001.
- [19] D. Brockmann, L. Hufnagel, and T. Geisel, "The scaling laws of human travel," *Nature*, vol. 439, no. 7075, pp. 462–465, Jan. 2006, doi: 10.1038/nature04292.
- [20] C. Saloma, G. J. Perez, G. Tapang, M. Lim, and C. Palmes-Saloma, "Selforganized queuing and scale-free behavior in real escape panic," *Proc. Nat. Acad. Sci. USA*, vol. 100, no. 21, pp. 11947–11952, Oct. 2003, doi: 10.1073/pnas.2031912100.
- [21] J. Watts, R. E. Morss, C. M. Barton, and J. L. Demuth, "Conceptualizing and implementing an agent-based model of information flow and decision making during hurricane threats," *Environ. Model. Softw.*, vol. 122, Dec. 2019, Art. no. 104524, doi: 10.1016/j.envsoft.2019.104524.
- [22] T. Vicsek, A. Czirók, I. J. Farkas, and D. Helbing, "Application of statistical mechanics to collective motion in biology," *Phys. A, Stat. Mech. Appl.*, vol. 274, nos. 1–2, pp. 182–189, Dec. 1999, doi: 10.1016/S0378-4371(99)00317-9.

- [23] J. M. Watts, Jr., "Editorial: Rescuing truth from familiarity," *Fire Technol.*, vol. 36, no. 1, pp. 1–2, Feb. 2000, doi: 10.1023/A:1015492916791.
- [24] F. Haghpanah, B. W. Schafer, and S. Castro, "Application of bug navigation algorithms for large-scale agent-based evacuation modeling to support decision making," *Fire Saf. J.*, vol. 122, Jun. 2021, Art. no. 103322, doi: 10.1016/j.firesaf.2021.103322.
- [25] D. Frenkel and B. Smit, Understanding Molecular Simulation: From Algorithms to Applications, 2nd ed. San Diego, CA, USA: Academic, 2002.
- [26] D. C. Rapaport, *The Art of Molecular Dynamics Simulation*, 2nd ed. Cambridge, U.K.: Cambridge Univ. Press, 2004.
- [27] A. Seyfried, B. Steffen, and T. Lippert, "Basics of modelling the pedestrian flow," *Phys. A, Stat. Mech. Appl.*, vol. 368, no. 1, pp. 232–238, Aug. 2006, doi: 10.1016/j.physa.2005.11.052.
- [28] S. T. Rassia and C. I. Siettos, "Escape dynamics in office buildings: Using molecular dynamics to quantify the impact of certain aspects of human behavior during emergency evacuation," *Environ. Model. Assessment*, vol. 15, no. 5, pp. 411–418, Oct. 2010, doi: 10.1007/s10666-009-9209-3.
- [29] B. Barnes, S. Dunn, C. Pearson, and S. Wilkinson, "Improving human behaviour in macroscale city evacuation agent-based simulation," *Int. J. Disaster Risk Reduction*, vol. 60, Jun. 2021, Art. no. 102289, doi: 10.1016/j.ijdrr.2021.102289.
- [30] C. Delcea and L.-A. Cotfas, "Increasing awareness in classroom evacuation situations using agent-based modeling," *Phys. A, Stat. Mech. Appl.*, vol. 523, pp. 1400–1418, Jun. 2019, doi: 10.1016/j.physa.2019.04.137.
- [31] D. Biner and N. Brun, "Evacuation bottleneck: Simulation and analysis of an evacuation of a lecture room with MATLAB," Eth Zurich, Zurich, Switzerland, Tech. Rep., 2018, p. 69. [Online]. Available: https://ethz.ch/en.html
- [32] F. E. Cornes, G. A. Frank, and C. O. Dorso, "Microscopic dynamics of the evacuation phenomena in the context of the social force model," *Phys. A, Stat. Mech. Appl.*, vol. 568, Apr. 2021, Art. no. 125744, doi: 10.1016/j.physa.2021.125744.
- [33] Y. Sun and H. Liu, "Crowd evacuation simulation method combining the density field and social force model," *Phys. A, Stat. Mech. Appl.*, vol. 566, Mar. 2021, Art. no. 125652, doi: 10.1016/j.physa.2020.125652.
- [34] X. Yang, X. Yang, Y. Li, J. Zhang, and Y. Kang, "Obstacle avoidance in the improved social force model based on ant colony optimization during pedestrian evacuation," *Phys. A, Stat. Mech. Appl.*, vol. 583, Dec. 2021, Art. no. 126256, doi: 10.1016/j.physa.2021.126256.
- [35] J. M. Epstein, "Agent-based computational models and generative social science," *Complexity*, vol. 4, no. 5, pp. 41–60, May 1999.
- [36] C. Delcea, L.-A. Cotfas, I.-A. Bradea, M.-I. Boloş, and G. Ferruzzi, "Investigating the exits' symmetry impact on the evacuation process of classrooms and lecture Halls: An agent-based modeling approach," *Symmetry*, vol. 12, no. 4, p. 627, Apr. 2020, doi: 10.3390/sym12040627.
- [37] X. Zheng, W. Li, and C. Guan, "Simulation of evacuation processes in a square with a partition wall using a cellular automaton model for pedestrian dynamics," *Phys. A, Stat. Mech. Appl.*, vol. 389, no. 11, pp. 2177–2188, Jun. 2010, doi: 10.1016/j.physa.2010.01.048.
- [38] D. Helbing, L. Buzna, A. Johansson, and T. Werner, "Self-organized pedestrian crowd dynamics: Experiments, simulations, and design solutions," *Transp. Sci.*, vol. 39, no. 1, pp. 1–24, Feb. 2005, doi: 10.1287/trsc.1040.0108.
- [39] J. C. González-Avella, V. M. Eguíluz, M. G. Cosenza, K. Klemm, J. L. Herrera, and M. S. Miguel, "Local versus global interactions in nonequilibrium transitions: A model of social dynamics," *Phys. Rev. E, Stat. Phys. Plasmas Fluids Relat. Interdiscip. Top.*, vol. 73, no. 4, Apr. 2006, Art. no. 046119, doi: 10.1103/PhysRevE.73.046119.
- [40] R. Liu, D. Jiang, and L. Shi, "Agent-based simulation of alternative classroom evacuation scenarios," *Frontiers Archit. Res.*, vol. 5, no. 1, pp. 111–125, Mar. 2016, doi: 10.1016/j.foar.2015.12.002.
- [41] Q. Zou, D. S. Fernandes, and S. Chen, "Agent-based evacuation simulation from subway train and platform," *J. Transp. Saf. Secur.*, vol. 13, no. 3, pp. 318–339, Jul. 2019, doi: 10.1080/19439962.2019.1634661.
- [42] N. Wagner and V. Agrawal, "An agent-based simulation system for concert venue crowd evacuation modeling in the presence of a fire disaster," *Expert Syst. Appl.*, vol. 41, no. 6, pp. 2807–2815, May 2014, doi: 10.1016/j.eswa.2013.10.013.
- [43] P. Dang, J. Zhu, S. Pirasteh, W. Li, J. You, B. Xu, and C. Liang, "A chain navigation grid based on cellular automata for large-scale crowd evacuation in virtual reality," *Int. J. Appl. Earth Observ. Geoinf.*, vol. 103, Dec. 2021, Art. no. 102507, doi: 10.1016/j.jag.2021. 102507.

- [44] A. Abdelghany, K. Abdelghany, H. Mahmassani, and W. Alhalabi, "Modeling framework for optimal evacuation of large-scale crowded pedestrian facilities," *Eur. J. Oper. Res.*, vol. 237, no. 3, pp. 1105–1118, Sep. 2014, doi: 10.1016/j.ejor.2014.02.054.
- [45] E. Ronchi, F. N. Uriz, X. Criel, and P. Reilly, "Modelling large-scale evacuation of music festivals," *Case Stud. Fire Saf.*, vol. 5, pp. 11–19, May 2016, doi: 10.1016/j.csfs.2015.12.002.
- [46] J. Shi, A. Ren, and C. Chen, "Agent-based evacuation model of large public buildings under fire conditions," *Automat. Construct.*, vol. 18, no. 3, pp. 338–347, May 2009, doi: 10.1016/j.autcon.2008.09.009.
- [47] Y. Wu, J. Kang, and C. Wang, "A crowd route choice evacuation model in large indoor building spaces," *Frontiers Archit. Res.*, vol. 7, no. 2, pp. 135–150, Jun. 2018, doi: 10.1016/j.foar.2018.03.003.
- [48] R. Zhou, Y. Ou, W. Tang, Q. Wang, and B. Yu, "An emergency evacuation behavior simulation method combines personality traits and emotion contagion," *IEEE Access*, vol. 8, pp. 66693–66706, 2020, doi: 10.1109/ACCESS.2020.2985987.
- [49] C. Ren, C. Yang, and S. Jin, "Agent-based modeling and simulation on emergency evacuation," in *Complex Sciences*, vol. 5, J. Zhou, Ed. Berlin, Germany: Springer, 2009, pp. 1451–1461, doi: 10.1007/978-3-642-02469-6_25.
- [50] C. Delcea, L.-A. Cotfas, L. Craciun, and A. G. Molanescu, "Establishing the proper seating arrangement in elevated lecture Halls for a faster evacuation process," *IEEE Access*, vol. 7, pp. 48500–48513, 2019, doi: 10.1109/ACCESS.2019.2909637.
- [51] N. Siyam, O. Alqaryouti, and S. Abdallah, "Research issues in agentbased simulation for pedestrians evacuation," *IEEE Access*, vol. 8, pp. 134435–134455, 2020, doi: 10.1109/ACCESS.2019.2956880.
- [52] H. Sharbini, R. Sallehuddin, and H. Haron, "Crowd evacuation simulation model with soft computing optimization techniques: A systematic literature review," *J. Manage. Anal.*, vol. 8, no. 3, pp. 443–485, Jul. 2021, doi: 10.1080/23270012.2021.1881924.
- [53] V. Bayram, "Optimization models for large scale network evacuation planning and management: A literature review," *Surv. Oper. Res. Manage. Sci.*, vol. 21, no. 2, pp. 63–84, Dec. 2016, doi: 10.1016/j.sorms.2016. 11.001.
- [54] J. Chen, T. Shi, and N. Li, "Pedestrian evacuation simulation in indoor emergency situations: Approaches, models and tools," *Saf. Sci.*, vol. 142, Oct. 2021, Art. no. 105378, doi: 10.1016/j.ssci.2021. 105378.
- [55] C. Ponsiglione, V. Roma, F. Zampella, and G. Zollo, "The fairness/efficiency issue explored through El Farol bar model," in *Scientific Methods for the Treatment of Uncertainty in Social Sciences*, vol. 377, J. Gil-Aluja, A. Terceño-Gómez, J. C. Ferrer-Comalat, J. M. Merigó-Lindahl, and S. Linares-Mustarós, Eds. Cham, Switzerland: Springer, 2015, pp. 309–327, doi: 10.1007/978-3-319-19704-3_26.
- [56] F. Stonedahl and U. Wilensky, "Finding forms of flocking: Evolutionary search in ABM parameter-spaces," in *Multi-Agent-Based Simulation XI*, vol. 6532, T. Bosse, A. Geller, and C. M. Jonker, Eds. Berlin, Germany: Springer, 2011, pp. 61–75, doi: 10.1007/978-3-642-18345-4_5.
- [57] C. Delcea and I. A. Bradea, *Economic Cybernetics. An Equation-Based Modeling and Agent-Based Modeling Approach*. Craiova, Romania: Editura Universitara, 2017.
- [58] R. J. Milne, L.-A. Cotfas, and C. Delcea, "Minimizing health risks as a function of the number of airplane boarding groups," *Transportmetrica B, Transp. Dyn.*, vol. 10, no. 1, pp. 901–922, Aug. 2021, doi: 10.1080/21680566.2021.1968322.
- [59] H. Faroqi and M.-S. Mesgari, "Agent-based crowd simulation considering emotion contagion for emergency evacuation problem," *Int. Arch. Photogramm. Remote Sens. Spatial Inf. Sci.*, vol. XL-1W5, pp. 193–196, Dec. 2015, doi: 10.5194/isprsarchives-XL-1-W5-193-2015.
- [60] J. Chhatwal and T. He, "Economic evaluations with agent-based modelling: An introduction," *PharmacoEconomics*, vol. 33, no. 5, pp. 423–433, Jan. 2015, doi: 10.1007/s40273-015-0254-2.
- [61] V. Grimm, E. Revilla, U. Berger, F. Jeltsch, W. M. Mooij, S. F. Railsback, H.-H. Thulke, J. Weiner, T. Wiegand, and D. L. DeAngelis, "Patternoriented modeling of agent-based complex systems: Lessons from ecology," *Science*, vol. 310, no. 5750, pp. 987–991, Nov. 2005, doi: 10.1126/science.1116681.
- [62] C. Watkins, D. Massey, J. Brooks, K. Ross, and M. L. Zellner, "Understanding the mechanisms of collective decision making in ecological restoration: An agent-based model of actors and organizations," *Ecol. Soc.*, vol. 18, no. 2, p. 32, 2013, doi: 10.5751/ES-05497-180232.

- [63] J. Badham. (2015). Review of An Introduction to Agent-Based Modeling: Modeling Natural, Social, and Engineered Complex Systems With NetLogo. Accessed: Feb. 2, 2020. [Online]. Available: http://jasss.soc.surrey.ac.uk/18/4/reviews/2.html
- [64] U. Wilensky and W. Rand, An Introduction to Agent-Based Modeling: Modeling Natural, Social, and Engineered Complex Systems With NetLogo. Cambridge, MA, USA: MIT Press, 2015.
- [65] R. Lovreglio, E. Ronchi, and D. Nilsson, "A model of the decisionmaking process during pre-evacuation," *Fire Saf. J.*, vol. 78, pp. 168–179, Nov. 2015, doi: 10.1016/j.firesaf.2015.07.001.
- [66] L. Fu, J. Luo, M. Deng, L. Kong, and H. Kuang, "Simulation of evacuation processes in a large classroom using an improved cellular automaton model for pedestrian dynamics," *Proc. Eng.*, vol. 31, pp. 1066–1071, Jan. 2012, doi: 10.1016/j.proeng.2012.01.1143.
- [67] A. Edrisi, B. Lahoorpoor, and R. Lovreglio, "Simulating metro station evacuation using three agent-based exit choice models," *Case Stud. Transp. Policy*, vol. 9, no. 3, pp. 1261–1272, Sep. 2021, doi: 10.1016/j.cstp.2021.06.011.
- [68] T. Korhonen and S. Hostikka, "Fire dynamics simulator with evacuation FDS+Evac, version 5. Technical reference and user's guide," VTT Tech. Res. Centre Finland, Espoo, Finland, Tech. Rep. VTT-WORK-119, 2008.
- [69] M. Hashemi, "Emergency evacuation of people with disabilities: A survey of drills, simulations, and accessibility," *Cogent Eng.*, vol. 5, no. 1, Jan. 2018, Art. no. 1506304, doi: 10.1080/23311916.2018.1506304.
- [70] M. Haghani and M. Sarvi, "Human exit choice in crowded built environments: Investigating underlying behavioural differences between normal egress and emergency evacuations," *Fire Saf. J.*, vol. 85, pp. 1–9, Oct. 2016, doi: 10.1016/j.firesaf.2016.07.003.



LIVIU-ADRIAN COTFAS received the Ph.D. degree in economic informatics from the Bucharest University of Economic Studies, Bucharest, Romania.

In 2018, he has been a Visiting Professor at the Université de Technologie Belfort-Montbéliard, France. He is currently with the Economic Cybernetics and Informatics Department, Bucharest University of Economic Studies. His research interests include semantic web, agent-based mod-

eling, social media analysis, sentiment analysis, recommender systems, geographic information systems, grey systems theory, and artificial intelligence systems.

Dr. Cotfas is an Active Member of the Grey Uncertainty Analysis Association and the INFOREC Association. He has received several research awards, including the "Georgescu Roegen" Award for Excellent Scientific Research.



CAMELIA DELCEA received the Ph.D. degree in economic cybernetics and statistics from the Bucharest University of Economic Studies, Bucharest, Romania.

She is currently with the Economic Cybernetics and Informatics Department, Bucharest University of Economic Studies. She is an Active Member in the area of grey systems theory, agentbased modeling and sentiment analysis. She is an Active Member of the Grey Uncertainty Analysis

Association. Since 2009, she has obtained 25 international and national awards ("Best Paper Award," "Georgescu Roegen" for Excellent Scientific Research, "Excellent Paper Award," and Top Reviewers) and has been invited to deliver a keynote speech on grey systems themes in 2013, 2016, and 2017 at the IEEE GSIS Conference and in 2018 at the GSUA Conference.



LIVIA-DIANA IANCU is currently pursuing the Ph.D. degree in the area of agent-based modeling applied in public spaces evacuation, being interested in economic modeling and cybernetics applied to real-life situations.

She is also with the Economic Cybernetics and Informatics Department, Bucharest University of Economic Studies, Bucharest, Romania. She teaches seminars of basics of operations research and basics of economic cybernetics and macroeco-

nomics at the Faculty of Economic Cybernetics, Statistics and Informatics, Bucharest University of Economic Studies.



CORINA IOANĂŞ received the Ph.D. degree in economic studies from the Bucharest University of Economic Studies, Bucharest, Romania.

She is currently with the Accounting and Audit Department, Bucharest University of Economic Studies. Over the last years, she has been a member in eight national and international projects, having ten books written in the economic field. Her main research interests include accounting, audit, consumer behavior, economic modeling, economic policies, and risk analysis.

Dr. Ioanăş is an Active Member of the Centre for the Accounting and Management Information Systems, ARACIS (Romanian Agency for Ensuring the Higher Education Quality), and CECCAR (The Body of Expert and Licensed Accountants of Romania).



CRISTINA PONSIGLIONE is currently an Associate Professor at the Department of Industrial Engineering (DII), University of Naples Federico II. She is also an Associate Researcher with the CRESS, University of Surrey, U.K., and a member of the Centre for Management and Innovation in Healthcare (CIRMIS), University of Naples Federico II. She is the Reference Person of the Third Mission of the Department of Industrial Engineering, University of Naples Federico II, and the

Co-ordinator of the Business and Managerial Engineering Bachelor Program at the DII. The scientific activity carried out can be traced back to the following areas: organizational learning and innovation in business organizations and territorial systems, human resources and competences, organizational processes, decision-making processes, and performance in organizations in the health sector.

...